CSCE 330 Fall 2002 Quiz 14 Assigned Friday, 02/10/18

An activation record in the C2 language contains a return address and storage for local variables. What other address is always present in an activation record in the C3 language? Is this an address in code memory or data memory? Answer: The address of the beginning of the caller’s activation record, also known as the dynamic link. This is an address in data memory. (Note that, on the other hand, the return address is an address in code memory.)